\documentclass[12pt]{book}

\usepackage{enumerate}
\usepackage{calculus}
\pagestyle{empty}
\setlength{\parindent}{0pt}
\everymath{\displaystyle}
\newcommand{\answerspace}{\vspace{1.0in}}
\newcommand{\problemspace}{\bigskip}

\begin{document}

{\bf Math 165: Honors Calculus I}
\hfill\makebox[2.5in][r]{Name:\hrulefill}

{\bf Quiz 2}\quad {\sl Sept. 7, 1995}

\begin{enumerate}

\problemspace
\item Find all real numbers $x$ satisfying $|x+2| < |2x+2|$.
\answerspace
\answerspace

\problemspace
\item Find all polynomials $p$ of degree $\le 2$ satisfying $p(1/x) = 1/p(x)$
for all $x\ne 0$.

\newpage
\item
  \begin{enumerate}[a)]
  \item Define $n\choose k$.
  \answerspace

  \item State the Binomial Theorem.
  \answerspace

  \item Determine the coefficient of $x^{6}$ in $(3x^2 - 5)^{12}$
  (give its prime factorization!).
  \answerspace

  \end{enumerate}

\end{enumerate}

\end{document}
